Author’s response to reviews

Title: Consistency of Supplied Food and Dentition Status of the Elderly in Residential Care Homes

Authors:
Yoshiaki Nomura (nomura-y@tsurumi-u.ac.jp)
Ayako Okada (okada-a@tsurumi-u.ac.jp)
Ryoko Otsuka (otsuka-ryoko@tsurumi-u.ac.jp)
Kaoru Sogabe (sogabe-k@tsurumi-u.ac.jp)
Kiyoshige Yamane (kiyoshige1392@gmail.com)
Taku Yamamoto (gtfgene@gmail.com)
Yuko Shigeta (shigeta-y@tsurumi-u.ac.jp)
Shuji Shigemoto (shigemoto-s@tsurumi-u.ac.jp)
Takumi Ogawa (ogawa-t@tsurumi-u.ac.jp)
Nobuhiro Hanada (hanada-n@tsurumi-u.ac.jp)

Version: 1 Date: 11 Mar 2019

Author’s response to reviews:

Dear Alpdogan Kantarci section editor of BMC oral health

Thank you for your advice to improve out manuscript.
Your advise were:
- The discussion is cryptic and for non-experts like myself, does not provide a “take-home message”
- Highlighting what data in future would be useful to obtain. Also please provide a 'take-home message' for the discussion.

According to your advice, we expand the discussion. As the editorial office advised not to include track changes or highlighting the manuscript. The high lightened Discussion were placed this response letter.
High lightened

Yellow: Newly revised sentence. These are mainly described the contents of reference in detail.

Light blue: What this work tells

Magenta: what data in future would be useful to obtain

Green: Take-home message: Dentist should be participated in the discussion when deciding the supplied food consistency along with caregivers, nutritionists or dieticians.